Author's response to reviews

Title: Automatic diagnosis of melanoma using machine learning methods on a spectroscopic system

Authors:

Lin Li (linli6008@gmail.com)

Version: 2 Date: 27 May 2014

Author's response to reviews: see over
Dear Editors of BMC-Medical Informatics & Decision Making,

Thank you for providing comments on our initial submission. We have made changes in the manuscript based on your feedback.

1. The author names/email addresses in the submission system are now consistent with those in the manuscript.
2. The study was approved by the Institutional Review Board (IRB). Patients have consented to participate in the study and the study has not used patient identifiable information. We also added these statements in the manuscript (in the section “Methods”, on page 7, from line 153 to line 155).
3. Line numbers have been inserted in our manuscript.

Please address all correspondence concerning this manuscript to me at linli6008@gmail.com.

Thank you for your consideration of this manuscript.

Sincerely,

Lin Li